



REVIEW HISTORY

Paper: **“Cultural Theory from Dichotomy to Interface: Travelling Theory and Saidian Dialectics”**

Corresponding Author: Mohamed Hamoud Kassim Al-Mahfedi

Email: [mahfadi76@yahoo.com](mailto:mahfadi76@yahoo.com)

Doi: 10.19044/llc.v6no4a1

Peer review:

Reviewer 1: R. Samuel Thorpe

Oral Roberts University, Tulsa, USA

Reviewer 2: Yahya Saleh Hasan Dahami

Al Baha University, Saudi Arabia

Published: 29.12.2019

